Author's response to reviews

Title: COPD Uncovered: An International survey on the impact of chronic obstructive pulmonary disease (COPD) on a working age population

Authors:

Monica J Fletcher Ms (m.fletcher@educationforhealth.org)
Jane Upton Dr (j.upton@bham.ac.uk)
Judith Taylor-Fishwick Ms (tayloric@evms.edu)
Sonia A Buist Professor (buists@ohsu.edu)
Christine Jenkins Professor (cj@med.usyd.edu.au)
John Hutton Professor (jh602@york.ac.uk)
Neil Barnes Professor (neil.barnes@bartsandthelondon.nhs.uk)
Thys Van Der Molen Professor (T.van.der.Molen@med.umcg.nl)
John Walsh Professor (jwwalsh@alphaone.org)
Paul Jones Professor (pjones@sghms.ac.uk)
Samantha Walker Dr (swalker@asthma.org.uk)

Version: 5 Date: 1 July 2011

Author's response to reviews:

Please may we request that Figures be cropped on our behalf in order to reduce the white space surrounding them.